Author’s response to reviews

Title: Oncology Nurses’ Perceptions of Obstacles and Role at the End-of-Life Care: Cross Sectional Survey

Authors:

Aurelija Blazeviciene (aurelija.blazeviciene@gmail.com)
Jamesetta Newland (jan7@nuy.edu)
Vilija Čivinskienė (vilijacivinskiene@gmail.com)
Renea Beckstrand (renea@byu.edu)

Version: 2 Date: 19 Sep 2017

Author’s response to reviews:

September 18, 2017

Re: Manuscript ID: PCAR-D-17-00015R1
Oncology Nurses’ Perceptions Of Obstacles And Role At The End-Of-Life Care: Cross Sectional Survey”

Dear Dr. Matteo Pasini,

We appreciate your invitation to respond to the editor’s and reviewers’ comments to our manuscript, and we are pleased for an opportunity to revise to improve the paper. The reviewers’ notes were very helpful in clarifying some details of our study. We have summarized the reviewers’ comments and our responses to each point in the attached table.

Additionally, we have carefully followed your manuscript preparation guidelines for formatting.

We believe that the paper has now been much improved, and both methods and the key messages have been clarified in response to the constructive suggestions of the reviewers. Thank you for considering this revised version for publication.

Yours sincerely,

Aurelija Blazevicienė (on behalf of the all authors)
Assoc. Prof., Head of Nursing and Care Department
Faculty of Nursing, Medical Academy
Lithuanian University of Health Sciences
Eivenių st. 2, Kaunas
LT-44307, Lithuania
aurelija.blazeviciene@gmail.com
Tel. +370 682 45938